Author’s response to reviews

Title: Contextual socioeconomic determinants of cardiovascular risk factors in rural south-west China: A multilevel analysis

Authors:

Le Cai (caile002@hotmail.com)
Virasakdi Chongsuvivatwong (cvirasak@medicine.psu.ac.th)
Alan Geater (alan.g@psu.ac.th)

Version: 5 Date: 4 May 2007

Author’s response to reviews: see over
Cover Letter

To editors of BMC Public Health

Dear Sir or Madam:

I am submitting the revised manuscript “Contextual socioeconomic determinants of cardiovascular risk factors in rural south-west China: A multilevel analysis” (MS: 1713454718116369) to BMC Public Health. I have addressed my responses to the comments in the revised manuscript.

Best wishes.

Cai Le

191 Western Renmin Road, Department of Health Information and Economics, Faculty of Public Health, Kunming Medical College, Kunming 650031

e-mail: caile002@hotmail.com